THE SCOPE OF THE NON-DELEGABLE DUTY OF CARE IN 'A COHERENT LAW OF NEGLIGENCE' UNDER AUSTRALIAN LAW

ALLISON SILINK*

The appeal in AA v Trustees of the Roman Catholic Church for the Diocese of Maitland-Newcastle presents a long-awaited opportunity for the High Court of Australia to reconsider whether a non-delegable duty of care can be breached by intentionally inflicted harm, such as child sexual abuse. This argument was rejected in New South Wales v Lepore (2003) 212 CLR 511. In Lepore, the majority held that such wrongdoing constituted an intentional tort or a crime, but was not appropriately characterised as a breach of the defendant's non-delegable duty of care, as 'it involves more'. This appeal will revisit the nature of a non-delegable duty and its relationship to negligence. This article argues that a non-delegable duty of care is not a stand-alone tort, amenable to bespoke development, but a duty of care in negligence. Accordingly, the expansion of the non-delegable duty to respond to intentionally inflicted harm engages a broader question about the scope of negligence and whether it is a tort of unintentional injury or whether it is also breached by intentionally inflicted harm.

I	Introduction	2
II	Legal Issues in Institutional Abuse Claims	5
III	Facts and Procedural History in AA	7
Α	At First Instance	7
В	The New South Wales Court of Appeal	8

^{*} Associate Professor, Faculty of Law, University of Technology Sydney. I would like to thank Jodi Gardner, Peter Handford, Tim Paine, Pam Stewart and Anita Stuhmcke for their generous comments on earlier drafts, and my particular thanks to Desmond Ryan for many long debates and discussions about these ideas. I gratefully acknowledge the research assistance of Grant Stollery. Finally, my sincere thanks to the anonymous reviewers whose comments were much appreciated and suggestions gratefully incorporated.

	The Scope of the Non-Delegable Duty of Care to Apply to Intentionally Cau	
A	The Nature of the Non-Delegable Duty	.10
В	Does Negligence Extend to Intentionally Caused Harm?	.13
V	Factors Relevant to Overturning Lepore	.33
VI	Conclusion	.35

I INTRODUCTION

Just over 30 years ago, in Burnie Port Authority v General Jones Pty Ltd ('Burnie Port Authority'), 1 a five member majority of the High Court spoke of the emergence of 'a coherent law of negligence to dominate the territory of tortious liability for unintentional injury to the person or property of another'. The non-delegable duty of care is a subset of duties of care owed in negligence which cannot be discharged by selection of an appropriately qualified contractor.³ In *New South Wales v Lepore* ('Lepore'),4 the majority found that a non-delegable duty could not be breached by intentionally caused harm so as to render a school authority liable for sexual abuse of a child by a teacher as it 'involves more' and to do so would 'sever the duty from its roots in the law of negligence'.6 In the appeal in AA v Trustees of the Roman Catholic Church for the Diocese of Maitland-Newcastle ('AA'),7 the Court will be invited to overturn *Lepore* and recognise that the non-delegable duty of care can be breached by intentionally inflicted harm. This raises broader questions as to whether the law of negligence more generally should remain 'the territory of tortious liability for unintentional injury to the person or property of another's or whether it can already, or should be developed to, apply to intentionally caused harm.

¹ (1994) 179 CLR 520 ('Burnie Port Authority').

² Ibid 544 (Mason CJ, Deane, Dawson, Toohey and Gaudron JJ).

³ Pafburn Pty Ltd v The Owners – Strata Plan No 84674 (2024) 99 ALJR 148, 154 [20] (Gageler CJ, Gleeson, Jagot and Beech-Jones JJ) ('Pafburn').

⁴ (2003) 212 CLR 511 ('Lepore').

⁵ Ibid 532 [31] (Gleeson CJ, Callinan J agreeing at 624 [339]).

⁶ Ibid 602 [266] (Gummow and Hayne JJ). See also *Trustees of the Roman Catholic Church for the Diocese of Maitland-Newcastle v AA* [2025] NSWCA 72, [160] (Leeming JA) (*'Trustees of the Roman Catholic Church'*).

⁷ (High Court of Australia, S94/2025, commenced 17 June 2025) ('AA').

⁸ Burnie Port Authority (n 1) 544 (Mason CJ, Deane, Dawson, Toohey and Gaudron JJ).

Like *Lepore*, the appeal in *AA* concerns a claim of child sexual abuse in an institutional context, following *Bird v DP* (a *Pseudonym*) ('*Bird*')⁹ and other recent cases that have reached the High Court on different legal issues in the same context.¹⁰ There is a common thread to these cases. Since the 1980s and 1990s, revelations of the extent of historical and ongoing child sexual abuse occurring in institutions entrusted with the care of children have shocked the world, and continue to do so. However, the pathways for survivors to seek compensation from the institutions in whose care they were abused are often blocked by insurmountable doctrinal hurdles. If the institution has not been negligent itself, the question will be whether the institution can be found liable for the wrongdoing of the abuser under the doctrines of vicarious liability, or by the operation of a non-delegable duty of care owed to the plaintiff.

One of the apparent tensions in the law is that claims in respect of institutional child abuse that may be factually very similar can have very different outcomes, depending upon the classification of the working status of the perpetrator. If intentional wrongdoing is committed by an employee, then in certain circumstances, the institution can be vicariously liable because it is accepted that vicarious liability applies to intentional torts, like battery. However, if intentional wrongdoing is committed by an *independent contractor* hired by the institution, vicarious liability is unavailable. There is a bright line drawn around independent contractors, 11 who are supposed to carry their own insurance and manage their own liabilities. The non-delegable duty of care which arises in special circumstances can render an institution personally liable for the *negligence* of independent contractors, but not for their intentional wrongdoing. To the community of families attending a school, a childcare centre, a church or other association, the institutional workforce appears as one. But not in the eye of the law. Given the fortress-like wall around vicarious liability limiting its scope to employee wrongdoing, it is therefore understandable that attention is trained on the non-delegable duty of care, and whether it can be expanded in scope. However, as Giliker has noted, there is a very real question as to

⁹ (2024) 98 ALJR 1349 ('Bird').

¹⁰ Prince Alfred College Inc v ADC (2016) 258 CLR 134 ('Prince Alfred College'); GLJ v Trustees of the Roman Catholic Church for the Diocese of Lismore (2023) 97 ALJR 857; Willmot v Queensland (2024) 98 ALJR 1407; RC v The Salvation Army (Western Australia) Property Trust (2024) 98 ALJR 1453.

¹¹ See, eg, *CCIG Investments Pty Ltd v Schokman* (2023) 97 ALJR 551, 561–2 [51] (Edelman and Steward JJ) (*'Schokman'*); *Bird* (n 9) [48] (Gageler CJ, Gordon, Edelman, Steward and Beech-Jones JJ; Gleeson J contra at [100]).

whether recourse to the non-delegable duty 'derives from frustration at the limits of vicarious liability rather than any considered conceptual development'. ¹²

This article considers the issues in relation to the scope of the non-delegable duty of care to apply to intentionally caused harm. The first aim of this article is to clarify the current Australian law.¹³ It argues that the contemporary scope of the non-delegable duty of care is not uncertain. It shows that the non-delegable duty of care under Australian law is accepted as a duty of care owed in negligence, albeit one that cannot be discharged by selection of an appropriately qualified contractor. It is not a stand-alone duty in tort that has a substantively different nature, or that is capable of bespoke development. Nor is it an absolute duty. The cases confirm that it is a 'species' of negligence.¹⁴ The corollary is that its scope is determined by the law of negligence. If the non-delegable duty of care were developed as a standalone tort, it would require consideration of the effect that such a development would have on existing doctrines, such as vicarious liability, and the coherence of the law as a whole.¹⁵

The second aim of this article is to consider whether the Australian law of negligence remains concerned with 'unintentional injury to the person or property of another'. In doing so, the article is not seeking to argue what the law *ought* to be, rather, what it *is*. It is argued that, despite some inconsistency in the cases, and views of scholars to the contrary, the better view is that negligence is concerned with unintended injury and does not extend to intentionally inflicted injury. There is no doubt that negligence can arise from intentional *acts*, but not from intentionally inflicted harm. This distinction is crucial to understanding the present limitation on the scope of negligence, and by extension, the non-delegable duty. It is argued that this has consequences for the approach to be taken to the proposed extension of the non-delegable duty. In order to maintain coherence within the law of negligence, the non-delegable duty ought not to be expanded in isolation to have a scope inconsistent with the law of negligence; this would require accepting that the law of

¹² Paula Giliker, *Vicarious Liability in Tort: A Comparative Perspective* (Cambridge University Press, 2010) 144, cited in *Cox v Ministry of Justice* [2014] EWCA 132, [61] (Beatson LJ). See also Jonathan Morgan, 'Liability for Independent Contractors in Contract and Tort: Duties to Ensure that Care is Taken' (2015) 74(1) *Cambridge Law Journal* 109, 122.

¹³ In doing so, it endeavours to apply the approach described by Jane Stapleton as '[addressing] the law as it is' rather than seeking out a 'best fit' for a 'Grand Theory'. See Jane Stapleton, *Three Essays on Torts* (Oxford University Press, 2021) 17.

¹⁴ See also John Murphy, 'The Liability Bases of Common Law Non-Delegable Duties: A Reply to Christian Witting' (2007) 30(1) *University of New South Wales Law Journal* 86.

¹⁵ Perera v Genworth Financial Mortgage Insurance Pty Ltd (2017) 94 NSWLR 83, 94 [42] (Leeming JA) ('Perera'), quoting Breen v Williams (1996) 186 CLR 71, 115 (Gaudron and McHugh JJ) ('Breen'). ¹⁶ Burnie Port Authority (n 1) 544 (Mason CJ, Deane, Dawson, Toohey and Gaudron JJ).

negligence *as a whole* can be breached by the forms of criminally intentional wrongdoing discussed in *Lepore*. If 'homicide, rape, and theft',¹⁷ as identified by Gleeson CJ, were also recognised forms of negligent conduct, that would be a very significant 'conquest in the imperial expansion of the law of negligence'.¹⁸

The article is structured as follows. The discussion commences in Part II by outlining the legal issues in relation to claims against institutional defendants in the context of institutional child abuse under Australian law. Part III summarises the facts and background to the appeal in AA. Part IV considers the scope of the non-delegable duty to apply to intentionally caused harm. Part V addresses factors that are likely to be relevant in seeking to overturn *Lepore*. Part VI concludes.

II LEGAL ISSUES IN INSTITUTIONAL ABUSE CLAIMS

The causes of action at common law typically relied upon in a claim against an institutional defendant will include one or more of negligence, vicarious liability or a non-delegable duty. However, in practice, there can be a range of obstacles to any action succeeding under Australian law. For example, an unincorporated association cannot be sued in its own name at common law as it has no legal identity. This limits claims against many defendants, particularly faith-based organisations, that have been historically structured as unincorporated associations. 19 There can be significant practical difficulties in establishing that a duty of care was owed at the time of the relevant conduct, particularly where the offence is alleged to have occurred decades ago and prior to the now widespread community awareness of the risks of child sexual abuse in an institutional context.²⁰ Whilst a defendant may be vicariously liable for the intentional wrongdoing of an employee, including sexual abuse,²¹ vicarious liability does not extend to independent contractors. Furthermore, the High Court has recently rejected the extension of vicarious liability to wrongdoing by persons 'akin to an employee'.22 This precludes vicarious liability in respect of abuse perpetrated by people working in other positions, particularly clergy, that are not traditional relationships of employment, even though vicarious liability under the common law in England and Wales has extended to such

¹⁷ Lepore (n 4) 532 [31].

¹⁸ Burnie Port Authority (n 1) 570 (Brennan J). Cf Astley v Austrust Ltd (1999) 197 CLR 1, 23 [48] (Gleeson CJ, McHugh, Gummow and Hayne JJ).

¹⁹ Trustees of the Roman Catholic Church v Ellis (2007) 70 NSWLR 565, 575 [47] (Mason P, Ipp JA agreeing at 603 [200], McColl JA agreeing at 603 [201]). This has been addressed by law reform: see below n 29.

²⁰ Trustees of the Roman Catholic Church (n 6) [201].

²¹ Prince Alfred College (n 10).

²² Bird (n 9).

relationships for over a decade.²³ The non-delegable duty of care applies to wrongdoing by independent contractors, but is not capable of being breached by intentionally caused harm,²⁴ thus precluding breach of a non-delegable duty arising from child sexual abuse.

Compensation from an institution in relation to the intentional wrongdoing of anyone other than a true employee therefore falls outside the scope of common law actions. The case law records the efforts that have been made through court proceedings to overcome these doctrinal boundaries under Australian law, all with little success.²⁵ State and territory legislatures have responded to certain obstacles in the pathways to justice for survivors through several rounds of targeted legislative reform to remove limitation periods, 26 and, in some jurisdictions, to reform civil liability legislation to create a statutory form of vicarious liability for institutional child sexual abuse (with prospective operation)²⁷ and to introduce new statutory duties of care.²⁸ Further reforms now allow for claims against unincorporated associations in respect of child abuse proceedings; for the appointment of a proper defendant to such claims brought against unincorporated associations; and for the recovery of damages from assets held on trust for an unincorporated association.²⁹ However, the limitations in the scope of vicarious liability and the non-delegable duty of care at common law remain insuperable obstacles for many survivors of historical institutional child sexual abuse.

²³ See, eg, Various Claimants v Catholic Child Welfare Society (Catholic Child Welfare) [2013] 2 AC 1; Cox v Ministry of Justice [2016] AC 660; Armes v Nottinghamshire County Council [2018] AC 355 ('Armes'); BXB v Trustees of the Barry Congregation of Jehovah's Witnesses [2024] AC 567.

Lepore (n 4) 531-33 [31]-[34].
 See, eg, Lepore v New South Wales (2001) 52 NSWLR

²⁵ See, eg, Lepore v New South Wales (2001) 52 NSWLR 420 ('Lepore (NSWCA)'); DP (a Pseudonym) v Bird [2021] VSC 850; AA v Trustees of the Roman Catholic Church for the Diocese of Maitland-Newcastle (2024) 334 IR 70 ('AA (NSWSC)').

 $^{^{26}}$ Limitation of Actions Act 1936 (SA) s 3A(1); Limitation of Actions Act 1958 (Vic) pt IIA div 5; Limitation Act 1969 (NSW) s 6A(1); Limitation of Actions Act 1974 (Qld) s 11A(1); Limitation Act 1974 (Tas) s 5B(1); Limitation Act 1981 (NT) s 5A(2); Limitation Act 1985 (ACT) s 21C(2); Limitation Act 2005 (WA) s 6A(2).

²⁷ Civil Liability Act 2002 (SA) pt 7A; Civil Liability Act 2002 (NSW) pt 1B; Civil Liability Act 2002 (Tas) pt 10C; Personal Injuries (Liabilities and Damages) Act 2003 (NT) pt 3A.

²⁸ Civil Liability Act 1936 (SA) pt 7A div 2; Wrongs Act 1958 (Vic) pt XIII; Civil Liability Act 2002 (NSW) pt 1B div 2; Civil Liability Act 2002 (Tas) pt 10C div 2; Personal Injuries (Liabilities and Damages) Act 2003 (NT) pt 3A div 2; Civil Liability Act 2003 (Qld) pt 2A div 2.

²⁹ Civil Liability Act 1936 (SA) s 50K; Wrongs Act 1958 (Vic) s 92(1); Civil Liability Act 2002 (NSW) pt 1B div 4 s 6N; Civil Law (Wrongs) Act 2002 (ACT) pt 8A.2; Civil Liability Act 2002 (Tas) pt 10C div 4; Civil Liability Act 2002 (WA) pt 2A div 2; Personal Injuries (Liabilities and Damages) Act 2003 (NT) pt 3A div 4; Civil Liability Act 2003 (Qld) pt 2A div 3.

The appeal in AA raises acutely the interaction of these common law obstacles and statutory complexities in the context of institutional child sexual abuse claims, particularly in relation to the liability of an unincorporated association. The proceedings in AA are discussed next.

III FACTS AND PROCEDURAL HISTORY IN AA

A At First Instance

The appellant in the appeal to the High Court was the plaintiff at first instance. He brought proceedings in relation to claims that he had been sexually abused in the 1960s by a priest, Father Ronald Pickin, in the presbytery of St Patrick's Catholic Church, Wallsend, in the Catholic diocese of Maitland-Newcastle, New South Wales. The Diocese of Maitland-Newcastle is an unincorporated association. The proceedings were commenced against the Trustees of the Roman Catholic Church for the Diocese of Maitland-Newcastle as the proper defendant under s 60 of the *Civil Liability Act 2002* (NSW) in respect of claims arising against the Diocese in negligence, vicarious liability and breach of a non-delegable duty (although the claims were not brought against the Diocese).

The assaults were alleged to have occurred when the plaintiff and a friend, Mr Perry, were in early high school and were invited by Father Pickin into the presbytery of the local church on Friday nights to consume alcohol, smoke cigarettes and play on a gambling machine in the bedroom when they were teenagers. The plaintiff claimed that Father Pickin committed penile-oral intercourse upon him, when the plaintiff was 'paralytic drunk', and they were alone in the presbytery together. Both Father Pickin and the Bishop of the Diocese had died before the litigation commenced. Mr Perry was called in the defence case. He agreed that he had visited the presbytery on Friday evenings with the plaintiff, where Father Pickin gave them alcohol and perhaps cigarettes, but said that other youths were present, and denied ever being sent out to buy cigarettes or having ever seen the plaintiff so drunk or assaulted.

The primary judge, Schmidt AJ, found the Diocese both vicariously liable³¹ and liable in negligence.³² The duty of care found by the court was based on a combination of the evidence led by the plaintiff about the awareness which bishops and other senior members of the Church then had about the risks which its priests could pose to children, the layout of the presbytery and lack of requirement to have other adults present which enabled Father Pickin to give the boys cigarettes and alcohol.

³⁰ AA (NSWSC) (n 25).

³¹ Ibid 105 [219].

³² Ibid 110 [264].

Together with evidence of the complaints which they had received, which was not widely known in the community, her Honour accepted that these facts together gave rise to a duty of care owed to *AA*. Despite some inconsistencies in the plaintiff's evidence, the primary judge accepted that the abuse occurred on the basis that the plaintiff's account was 'vivid'³³ and consistent with tendency evidence led that Father Pickin had touched other teenage male students' genitals. Her Honour found that the duty of care was breached and awarded damages of \$636,480. Her Honour did not determine the claim in relation to a non-delegable duty.

B The New South Wales Court of Appeal

The Trustees appealed, submitting that the primary judge erred in finding that the abuse occurred, and that the appellant owed the plaintiff a duty of care which had been breached. On appeal, it was accepted that the judgment below based on vicarious liability could not stand after the High Court decision in *Bird*. However, by a notice of contention, the plaintiff sought to uphold the judgment of the primary judge based on a non-delegable duty.

Leeming JA, with whose reasons Bell CJ agreed, found that the fact-finding process at trial had miscarried. His Honour found that the primary judge had relied heavily on the vividness of the plaintiff's account without adequately addressing the possibility that this represented 'a sincerely held but unreliable belief'. Leeming JA found that in a number of key respects the plaintiff's evidence was demonstrably unreliable.35 Leeming JA held that the notice of contention in relation to the non-delegable duty was precluded by the ratio of the High Court's decision in Lepore which had not been overturned. Despite questions as to whether the proceedings had been brought properly against the Trustees given that proceedings were not commenced against the Diocese, Leeming JA was prepared to assume that s 60 of the Civil Liability Act 2002 (NSW) applied to permit the proceedings to be brought against the nominated defendant (although his Honour queried whether the proceedings were still required to be commenced against the association). However, Leeming JA found that no duty of care was owed by the Diocese in 1969. It was held that the existence of a duty of care turned on establishing knowledge by the Bishop or senior members of the Diocese which the plaintiff had not done. Leeming JA concluded that, whilst it may be accepted that a bishop with power to appoint and remove a priest who knew that a priest had a history of sexual assault would be

-

³³ Ibid 97 [164].

³⁴ Trustees of the Roman Catholic Church (n 6) [131], [136], [145].

³⁵ Ibid [138].

under a duty to take reasonable steps to prevent further assaults,³⁶ it did not follow that in 1969 a reasonable person in the position of the Bishop would have taken steps 'on the basis that every priest was a potential child abuser'.³⁷ Leeming JA found that expert evidence about bishops' general awareness of risks was speculative and unsupported, and that a 1987 psychiatric report mentioning a 1954 complaint about another priest who had left the Diocese by 1969 was insufficient to establish relevant knowledge.³⁸

Ball JA disagreed that the primary judge had erred in concluding that the plaintiff was sexually abused.³⁹ However, his Honour accepted that no duty of care was owed in 1969.⁴⁰ In the result the appeal was allowed on the grounds that the fact-finding process miscarried; that the claim in vicarious liability was unsustainable and non-delegable duty precluded by authority; and that no duty of care was owed in 1969.

The three grounds of the appeal to the High Court by the plaintiff at trial focus on the non-delegable duty and the duty of care. The first ground of appeal seeks to overturn *Lepore* on the question of intentional wrongdoing and the scope of the non-delegable duty.⁴¹ The second ground is whether a Catholic diocese owed a non-delegable duty (of the nature for which the appellant contends) in the 1960s to children entrusted to the pastoral care of a priest for religious education so as to found a cause of action brought against a nominated defendant.⁴² The third ground relates to the level of knowledge on the part of a bishop required to give rise to a duty of care in the 1960s, and specifically whether it is necessary to show that 'the Bishop or senior priests in the diocese knew that the priest posed a risk to children'.⁴³

Against this background to the appeal, Parts IV and V of this article focus on the scope of the non-delegable duty to be breached by intentionally caused harm in the form of child sexual abuse which amounts to the tort of battery as well as a criminal offence.

³⁶ Ibid [217].

³⁷ Ibid [218].

³⁸ Ibid [213]-[217].

³⁹ Ibid [253].

⁴⁰ Ibid.

⁴¹ AA, 'Appellant's Submissions', Submission in *AA v the Trustees of the Roman Catholic Church for the Diocese of Maitland-Newcastle*, S94/2025, 8 July 2025, [24]–[47].

⁴² Ibid [48]–[55].

⁴³ Ibid [4].

IV THE SCOPE OF THE NON-DELEGABLE DUTY OF CARE TO APPLY TO INTENTIONALLY CAUSED HARM

The non-delegable duty of care had been identified by courts⁴⁴ and scholars⁴⁵ as a potential means to render institutional defendants liable for child sexual abuse. This part considers the nature of the non-delegable duty of care as a species of negligence and the consequences of that characterisation for its scope to be developed to apply to intentionally caused harm.

A The Nature of the Non-Delegable Duty

The nature of a common law non-delegable duty of care under Australian law was recently described by the majority of the High Court in *Pafburn Pty Ltd v The Owners – Strata Plan No 84674 ('Pafburn')*, 46 as follows:

Generally, a "non-delegable duty" is a type of duty of care which, if owed by a person, means that the person cannot exclude or limit their liability for conduct within the scope of the duty of care causing reasonably foreseeable harm merely by the person exercising reasonable care in arranging for another person to perform the function to which the non-delegable duty of care attaches. That is, although the function to which a non-delegable duty of care attaches is "delegable" in that the person subject to the duty may procure performance of any function within the scope of the duty by another person, the non-delegable duty is not satisfied merely by the taking of reasonable care by the person subject to the duty, because the content of the duty is personally to ensure that that other person performing the function in fact takes reasonable care.⁴⁷

⁴⁴ See, eg, Schokman (n 11) [81] (Edelman and Steward JJ).

⁴⁵ See, eg, Paula Giliker, 'Analysing Institutional Liability for Child Sexual Abuse in England and Wales and Australia: Vicarious Liability, Non-Delegable Duties and Statutory Intervention' (2018) 77(3) *Cambridge Law Journal* 506; Francis Santayana, 'Vicarious Liability, Non-Delegable Duties and the "Intentional Wrongdoing Problem"' (2019) 25(2) *Torts Law Journal* 152; Robert Stevens, 'Non-Delegable Duties and Vicarious Liability' in Jason Neyers, Erika Chamberlain and Stephen Pitel (eds), *Emerging Issues in Tort Law* (Bloomsbury, 2007) 361 ('Stevens'); Neil Foster, 'Convergence and Divergence: The Law of Non-Delegable Duties in Australia and the United Kingdom' in Andrew Robertson and Michael Tilbury (eds), *Divergences in Private Law* (Hart, 2016) 197; Christine Beuermann, *Reconceptualising Strict Liability for the Tort of Another* (Hart, 2019) 49. See also Jane Wangmann, 'Liability for Institutional Child Sexual Assault: Where Does Lepore Leave Australia?' (2004) 28(1) *Melbourne University Law Review* 169.

⁴⁶ *Pafburn* (n 3).

⁴⁷ Ibid 154 [20] (Gageler CJ, Gleeson, Jagot and Beech-Jones JJ), citing *Lepore* (n 4) 527–529 [20]–[21]; *Commonwealth v Introvigne* (1982) 150 CLR 258, 270 (Mason J) ('Introvigne'); *McDermid v Nash Dredging & Reclamation Co Ltd* [1987] AC 906, 910 (Lord Hailsham); *Northern Sandblasting Pty Ltd v Harris* (1997) 188 CLR 313, 350 (Toohey J); *Leichhardt Municipal Council v Montgomery* (2007) 230

Pafburn lends no support to the conception of the non-delegable duty of care as unrelated to the doctrine of negligence. In particular, the case does not support a characterisation of it as a standalone tort of absolute duty to prevent harm.⁴⁸ This is consistent with earlier cases such as Leichhardt Municipal Council,49 where Kirby J observed that '[t]he supposition of a non-delegable duty does not alter the content of the tort sued for, nor does it substitute a different and free-standing tort.'50 His Honour there said that it 'does no more than to affirm the imposition of personal liability on the duty-holder, which cannot be discharged, as otherwise it would, by selecting an apparently reputable and competent contractor'.51 In Burnie Port Authority,⁵² Brennan I discussed the suggestion that had been made that the non-delegable duty was 'an absolute duty 'to prevent...mischief',53 a duty higher than a duty to exercise reasonable care. However, his Honour noted that 'the duty in negligence is not so demanding'.54 In that case, the majority also noted that in Kondis v State Transport Authority,55 Mason J had 'identified some of the principal categories of case in which the duty to take reasonable care under the ordinary law of negligence is non-delegable in that sense'. 56 As these cases confirm, it is a 'species' of negligence, albeit one that cannot be discharged by selection of an appropriately qualified contractor.⁵⁷

The majority in *Pafburn* cited *Lepore* as one of the authorities for this description of the nature of the non-delegable duty of care.⁵⁸ The Court's reasons in *Lepore* are discussed in detail below, but for present purposes, Gummow and Hayne JJ expressly noted that all of the cases in which non-delegable duties have been considered by the High Court 'have been cases in which the plaintiff has been injured as a result of

CLR 22, 29 [9] (Gleeson CJ) ('Leichhardt Municipal Council'). As Denning LJ said in Cassidy v Ministry of Health [1951] 2 KB 343, 363: 'where a person is himself under a duty to use care, he cannot get rid of his responsibility by delegating the performance of it to someone else, no matter whether the delegation be to a servant under a contract of service or to an independent contractor under a contract for services.'

⁴⁸ Cf Christian Witting, 'Breach of the Non-Delegable Duty: Defending Limited Strict Liability in Tort' (2006) 29(3) *University of New South Wales Law Journal* 33.

⁴⁹ Leichhardt Municipal Council (n 47).

⁵⁰ Ibid 62 [109].

⁵¹ Ibid.

⁵² Burnie Port Authority (n 1).

⁵³ Ibid 576, quoting *Bower v Peate* (1876) 1 QBD 321, 326-7 (Cockburn CJ).

⁵⁴ Burnie Port Authority (n 1) 577.

⁵⁵ (1984) 154 CLR 672, 679–687.

⁵⁶ Burnie Port Authority (n 1) 550.

⁵⁷ See also John Murphy, 'The Liability Bases of Common Law Non-Delegable Duties: A Reply to Christian Witting' (2007) 30(1) *University of New South Wales Law Journal* 86.

⁵⁸ *Pafburn* (n 3) 154 [20], citing *Lepore* (n 4).

negligence'.⁵⁹ The majority also cited the Court's earlier decision in *Commonwealth v Introvigne* ('Introvigne').⁶⁰ Like Lepore, Introvigne was a case dealing with the non-delegable duty owed by a school authority but unlike Lepore, in that case it was found to have been breached. In Introvigne, Mason J there described the duty of care owed by the school in classical terms drawn from Carmarthenshire County Council v Lewis ('Carmarthenshire'),⁶¹ observing that the liability of a school authority in negligence for injury suffered by a pupil attending the school is not a purely vicarious liability. His Honour stated:

A school authority owes to its pupil a duty to ensure that reasonable care is taken of them whilst they are on the school premises during hours when the school is open for attendance...[The decision in *Carmarthenshire*] proceeds on the footing that the duty is not discharged by merely appointing competent teaching staff and leaving it to the staff to take appropriate steps for the care of the children. It is a duty to ensure that reasonable steps are taken for the safety of the children, a duty the performance of which cannot be delegated.⁶²

All of these discussions are consistent with the non-delegable duty as a duty owed in negligence, but one that cannot be discharged through delegation to a competent contractor.

Some cases describe the non-delegable duty as a duty 'to take positive steps' to protect a plaintiff from particular harm.⁶³ The question has been raised whether this gives the non-delegable duty of care a different scope, such that it is breached if a particular harm (like abuse) occurs. In *Lepore*, Gaudron J addressed this argument. Her Honour noted that positively expressed formulations of the non-delegable duty such as a 'duty to provide a safe system of work' upon an employer, do not amount to strict liability for *any* harm. Her Honour observed that:

There is a tendency to speak, in the case of an employer, of a duty to provide a safe system of work or, in the case of an education authority, a duty to provide a safe school environment, without acknowledging either that, in that context, "safe" means "free of a foreseeable risk of harm" or that the duty is a duty to take reasonable care. If the duty of an education

⁵⁹ Lepore (n 4) 599 [256].

⁶⁰ *Pafburn* (n 3) 154 [20], citing *Introvigne* (n 47).

^{61 [1955]} AC 549.

⁶² Ibid 269–70 (Mason J, Gibbs CJ agreeing at 260).

⁶³ See, eg, Lepore (n 4) 552–3 [104]–[105] (Gaudron J); Kondis (n 54) 686–7 (Mason J, Deane J agreeing at 694, Dawson J agreeing at 695); Introvigne (n 46) 271 (Mason J); Leichhardt Municipal Council (n 46) 64–5 [117]–[121] (Kirby J); Woodland v Swimming Teachers Association [2014] AC 537, 573 [7], 583 [23], 584–5 [25] (Lord Sumption JSC, Lords Clarke, Wilson and Toulson JJSC agreeing) ('Woodland').

authority to provide a safe school environment were not confined by considerations of foreseeability and reasonable care, it would result in strict liability in the sense that the authority would be liable upon proof of injury being sustained on school premises during school hours.⁶⁴

In other words, whilst the duty is commonly one that can be expressed positively,⁶⁵ that does not convert it into a positive duty to prevent all harm. Against this brief summary of the nature of the non-delegable duty as a duty of care owed in negligence under Australian law, the broader question is whether the law of negligence can be breached by intentionally caused harm.

B Does Negligence Extend to Intentionally Caused Harm?

This section addresses this foundational question that is anterior to the scope of non-delegable duties of care: does the law of negligence extend to intentionally caused harm, or 'deliberately and criminally inflicting injury',66 under Australian law? There are two competing views on the answer. It is argued here that the better view of High Court authority is that intentionally caused harm does not fall within the scope of the tort of negligence under the common law of Australia.⁶⁷ In *Burnie* Port Authority, the High Court noted that, from the historical action on the case, we can trace 'the subsequent emergence of a coherent law of negligence to dominate the territory of tortious liability for unintentional injury to the person or property of another'.68 This view recognises the historical form of action in moulding the development of negligence, but its contemporary form is shaped by precedent. Lepore remains the most detailed consideration of the question, where the majority clearly rejected the scope of the non-delegable duty of care, and by implication, the scope of negligence, to apply to intentionally inflicted personal injury.⁶⁹ Gleeson CJ, with whom Callinan J agreed, found that intentional criminality introduces a factor of legal relevance beyond a mere failure to take care in negligence. Gummow and Hayne JJ commented that all of the cases in which a non-delegable duty has been found have been cases of negligence, so to extend non-delegable duty to intentional acts would be to 'sever the duty from its roots in the law of negligence'. This is discussed in detail below and has been applied subsequently.

⁶⁴ Lepore (n 4) 552 [103] (Gaudron J).

⁶⁵ Ibid 552-3 [104] (Gaudron J).

⁶⁶ Ibid 531 [31] (Gleeson CJ).

⁶⁷ Ibid 531-2 [31] (Gleeson CJ).

⁶⁸ Burnie Port Authority (n 1) 544 (Mason CJ, Deane, Dawson, Toohey and Gaudron JJ).

⁶⁹ *Lepore* (n 4) 535 [38]–[39] (Gleeson CJ, Callinan J agreeing at 624 [339]), 601 [265] (Gummow and Hayne JJ).

⁷⁰ Ibid 602 [266] (Gummow and Hayne J).

The contrary view, for which there is stronger academic support and some judicial support, is that the modern law of negligence is conduct-based only and, therefore, whether harm is intentionally caused or not is irrelevant. Cane argues that '[n]egligence in tort law is failure to comply with a legally specified standard of conduct, pure and simple. It has no mental element.'71 On this view, arguably, '[h]omicide, rape, and theft'72 are acts that fall below the standard of conduct required and this is all that matters. Accordingly, as argued by Stevens, 'liability for deliberate abuse follows a fortiori from liability for want of care'.73 Handford comprehensively surveyed scholarly support for the proposition and also argued that 'the tort of negligence is a valuable addition to the armoury of weapons available for deliberate wrongdoing'. There are also judicial *dicta* suggesting that negligence can apply to intentionally caused harm. For example, in Schokman, Edelman and Steward II observed that 'it is at least arguable that the assault claim [in Lepore] could nevertheless have been pleaded as a breach of a duty of care'.75 In Carrier v Bonham, 76 McPherson IA suggested that '[w]hat we really have now is not two distinct torts of trespass and negligence, but a single tort of failing to use reasonable care to avoid damage however caused.'77

There is no question that *mens rea* is not an express element of negligence. From 19th century expressions of principle in cases such as *Heaven v Pender*,⁷⁸ in which Brett MR described 'actionable negligence' as 'the neglect of the use of ordinary care or skill towards a person to whom the defendant owes the duty of observing ordinary care and skill', through to Lord Atkin's famous dictum in *Donoghue v Stevenson*,⁷⁹ to modern formulations,⁸⁰ it is conspicuously absent and denied. As noted by the authors of the 11th edition of *Fleming's Law of Torts*:

⁷¹ Peter Cane, 'Mens Rea in Tort Law' (2000) 20(4) *Oxford Journal of Legal Studies* 535, 536. My thanks to the anonymous reviewer for reference to this article.

⁷² Lepore (n 4) 531–2 [31] (Gleeson CJ).

⁷³ Stevens (n 45) 361. See also Jack Maxwell, 'Liability of Educational Institutions for Child Abuse' (2019) 93(6) Australian Law Journal 477, 480, citing Robert Stevens, Torts and Rights (Oxford University Press, 2007) 271.

⁷⁴ Peter Handford, 'Intentional Negligence: A Contradiction in Terms?' (2010) 32(1) *Sydney Law Review* 29, 61.

⁷⁵ Schokman (n 11) 198 [78], citing Lepore (n 4); Williams v Milotin (1957) 97 CLR 465, 470–1 (Dixon CJ, McTiernan, Williams, Webb and Kitto JJ) ('Milotin'); Gray v Motor Accident Commission (1998) 196 CLR 1, 9–10 [22] (Gleeson CJ, McHugh, Gummow, and Hayne JJ), 28 [85] (Kirby J) ('Gray').

⁷⁶ [2002] 1 Qd R 474.

⁷⁷ Ibid 484 [27] (McPherson JA).

⁷⁸ (1883) 11 QBD 503, 509 ('Heaven').

⁷⁹ [1932] AC 562, 579-80 ('Donoghue').

⁸⁰ See, eg, Carolyn Sappideen et al, *Fleming's The Law of Torts* (Thomson Reuters, 11th ed, 2023) 141.

[I]t is important to grasp at the outset that negligence is not a state of mind, but conduct that falls below the standard regarded as normal or desirable. The subjective notion of personal 'fault' has long been discarded in favour of the stricter, impersonal standard of how a reasonable person should have acted in the circumstances.⁸¹

It is surprising that such a fundamental question has not been authoritatively settled. As discussed below, it is argued that the better view of the current law of Australia is that negligence does *not* apply to intentionally inflicted harm. However, it is not clear how this limitation arises, other than by judicial implication, or whether there remains a logical rationale for it. It is to be hoped that the High Court will clarify this foundational issue for the law of negligence, and in the process, clarify the question for the non-delegable duty.

1 The meaning of 'intention'

A brief preliminary exercise in ground clearing is necessary to address the meaning of 'intention' as it is used in relation to arguments that negligence does, or does not, extend to 'intentional conduct', 'intentional wrongdoing', and 'intentionally inflicted injury'. A comprehensive discussion of the role of intention in tort law is beyond the scope of this article, and this article is not intended to engage with various theoretical underpinnings for tortious liability, or the relevance of intention to any particular model.⁸² There are two narrow points to clarify for the limited purpose of understanding the different ways in which these terms are used in cases.

First, the word 'intentional' in the phrase an 'intentional act' is sometimes used simply to denote a *voluntary* act as opposed to an involuntary act.⁸³ There is no doubt that negligence can arise from an 'intentional act' in this sense. Voluntary acts are often conscious deliberate choices. I can choose to kick a ball. However, this does not imply an *intention to inflict harm* from that conduct.

Second, it is clear that negligence can arise from intentional conduct and even give rise to exemplary damages where the defendant is shown to have acted 'consciously in contumelious disregard of the rights of the plaintiff or persons in the position of the plaintiff'.⁸⁴ These are relatively rare cases in negligence. However, commonly

⁸² There have been a number of significant scholarly considerations of its role. See, eg, Cane (n 71); Handford (n 74); Peter Cane, 'Justice and Justifications for Tort Liability' (1982) 2(1) Oxford Journal of Legal Studies 30; FA Trindade, 'Intentional Torts: Some Thoughts on Assault and Battery' (1982) 2(2) Oxford Journal of Legal Studies 211; Philip Sales and Daniel Stilitz, 'Intentional Infliction of Harm by Unlawful Means' (1999) 115(Jul) Law Quarterly Review 411, 436–7.

⁸¹ Ibid.

⁸⁴ *Gray* (n 75) 9–10 [22] (Gleeson CJ, McHugh, Gummow and Hayne JJ).

cited examples where it has been awarded were still *unintentional* injury cases, rather than instances of an intention to cause harm. Exemplary damages have been awarded in employment cases, for example, where an employer has consciously allowed a risk of harm to continue on a worksite, such as the presence of asbestos, where the employer knew or ought to have known that it could cause harm. It has been held in some of these cases that the employer's decisions preferencing corporate profits were 'in contumelious disregard of the rights of the plaintiff' who later contracts a deadly lung disease or other physical illness. However, they did not arise from an intention to *cause* such harm. As observed by Mahoney P:

One may, I think, act 'in contumelious disregard of the plaintiff's rights' by doing recklessly and negligently what is likely to cause damage to the plaintiff even though it is not one's intention to cause it.⁸⁵

Accordingly, acting intentionally, even 'in contumelious disregard of the plaintiff's rights', is not to be equated with intentionally caused harm or 'intentional wrongdoing' and can arise in relation to unintentionally caused injuries within the law of negligence.

In some contexts, negligence is used to indicate 'inadvertence to consequences' as opposed to intention which signifies 'advertence to consequences'.⁸⁶ However, difficult questions arise in relation to reckless indifference as to a foreseeable outcome that was not desired, but not avoided. Cane has noted that recklessness can sometimes be elided with an intention to produce that harm.⁸⁷ It is beyond the scope of this article to explore in detail the role of recklessness in negligence. The focus here is on intentionally inflicted, direct physical contact in the form of child sexual abuse. In these cases, the perpetrator has intentionally committed this form of direct and unlawful harm that amounts to the tort of battery.

2 19th and 20th century developments

A brief sketch of the history of intentional and unintentional torts and their relationship to the forms of action in the nineteenth century provides context for the discussion of the development of negligence which follows. The point has been made that the distinction between the historical forms of action was originally between direct and indirect harm,⁸⁸ rather than between intentional and unintentional

⁸⁵ Trend Management Ltd v Borg (1996) 40 NSWLR 500, 504 (Mahoney P) ('Trend Management').

⁸⁶ Cane (n 82) 38.

⁸⁷ Ihid

⁸⁸ Scott v Shepherd (1773) 2 Black W 892; 96 ER 525, 527-8 (Blackstone J).

conduct. However, as noted by Cane,⁸⁹ gradually the distinction between negligence as inadvertent (careless) and unintentionally inflicted harm, on the one hand, and wilfully inflicted injury, on the other, became evident in the cases.

(a) Intentionally caused harm

Prior to the judicature reforms, choosing the appropriate form of action turned on whether the harm was direct or indirect. The presence of intention to cause the harm could be relevant to the appropriate cause of action. Directly caused harm was actionable in trespass, including if intentionally caused. These came to be known as the 'intentional' torts: of battery, assault and false imprisonment. Harm caused intentionally but *indirectly* came to be recognised in discrete actions on the case. For example, in *Bird v Holbrook* ('*Holbrook*'), he Court recognised that indirectly but intentionally caused physical injury was actionable on the case. *Holbrook* concerned the deliberate setting of a loaded gun, or 'spring gun', set in a garden with a trip wire 'for the express purpose of doing injury'. The teenage plaintiff was badly injured by a spring gun when he trespassed into the defendant's garden to retrieve his pea fowl which had strayed onto the defendant's property. In *Wilkinson v Downton*, he defendant was found liable for the plaintiff's injuries arising from severe shock following the defendant's conduct in deliberately telling the plaintiff a practical joke about her husband having been badly injured. Wright J found that:

The defendant has...wilfully done an act calculated to cause physical harm to the plaintiff — that is to say, to infringe her legal right to personal safety, and has in fact caused physical harm to her. That proposition without more appears to me to state a good cause of action, there being no justification for the act.⁹⁴

The conduct was characterised there as 'calculated to harm' because:

It is difficult to imagine that such a statement, made suddenly and with apparent seriousness, could fail to produce grave effects under the circumstances upon any but an exceptionally indifferent person, and therefore *an intention to produce such an effect must be imputed*...⁹⁵

⁸⁹ Cane (n 82) 37, citing *Ogle v Barnes* (1799) 8 TR 188; 101 ER 1338, *Williams v Holland* (1833) 6 Car & P 23; 172 ER 1129. Cane (n 82) observes (at 38) that '[t]hus it was, in the area of personal injuries, that the exclusive territory of Trespass came to be restricted to intentional, direct injuries.'

⁹⁰ See, eg, Trindade (n 82) for a discussion of these intentional torts.

^{91 (1828) 4} Bing 628; 130 ER 911 ('Holbrook').

⁹² Ibid 916 (Best CJ).

⁹³ [1897] 2 QB 57 ('*Downton*'). See generally Handford (n 74); M Lunney, 'Practical Joking and its Penalty: *Wilkinson v Downton* in Context' (2002) 10(3) *Tort Law Review* 168.

⁹⁴ Downton (n 93) 58-9.

⁹⁵ Ibid 59 (emphasis added).

Handford notes that in order to find for the plaintiff, Wright I had to create a new tort as no existing tort fitted the facts: it was treated as a cause of action arising from intentionally caused, indirect harm. Handford observed that '[t]he tort that he created is potentially quite wide-ranging: it covers any sort of conduct which is calculated to cause physical harm to the person, 96 and that a tort for intentionally inflicted, indirect harm could in theory apply to other intentionally caused, indirect harm cases, such as spring-gun cases like Bird v Holbrook, the administration of drugs or poison, the infliction of disease, and the well-known examples of logs on the highway.⁹⁷ However, its subsequent reach has been largely confined to physical harm by way of nervous shock. Different views have been taken as to whether it was properly a case of intentionally inflicted harm if the intention had to be imputed. As there was no intention to injure by telling the joke which caused the readily foreseeable harm, it has been suggested that the circumstances might be better analysed now under negligence.98

(b) Unintentionally caused harm

Williams v Holland99 was a significant nineteenth century case resolving that negligence was not confined by the existing forms of action, although both the parties and the court confined intentional wrongdoing (or wilful harm) to trespass and excluded it from the scope of negligence. Tindal CJ held that:

We think [Moreton v Hardern (1825) 4 B & C 223; 107 ER 1042] has laid down a plain and intelligible rule, that where the injury is occasioned by the carelessness and negligence of the Defendant, the Plaintiff is at liberty to bring an action on the case, notwithstanding the act is immediate, so long as it is not a wilful act; and, upon the authority of that case, we think the present form of action maintainable to recover damages for the iniurv.¹⁰⁰

⁹⁶ Peter Handford, 'Wilkinson v Downton and Acts Calculated to Cause Physical Harm' (1985) 16 University of Western Australia Law Review 31, 34.

⁹⁷ Ibid 35.

⁹⁸ The proper understanding of the case continues to be debated. There has been *obiter* in the High Court to suggest that the case might have been subsumed by negligence. See Magill v Magill (2006) 226 CLR 551, 589 [117] (Gummow, Kirby and Crennan JJ) ('Magill'). This was doubted by Spigelman CJ in Nationwide News Pty Ltd v Naidu (2007) 71 NSWLR 471, 486-7 [71]-[74] ('Nationwide News'). See also Scott Wotherspoon, 'Resuscitating the Wilkinson v Downton Tort in Australia' (2011) 85(1) Australian Law Journal 37.

^{99 (1833) 10} Bing 112; 131 ER 848 ('Williams').

¹⁰⁰ Ibid 850 (emphasis added). In *Moreton v Hardern* (1825) 4 B & C 223; 107 ER 1042, it was claimed that the defendants so carelessly managed their coach and horses, that the coach ran against the plaintiff and broke his leg. The jury found that it happened through his defendant's negligent driving.

For present purposes, it is the express exclusion of intentionally caused harm from the scope of negligence that is important. This was adopted under Australian law. *Williams* was applied in *Elliott v Barnes*. 101 Street CJ found that, '[i]f of course, the plaintiff were able to establish a wilful and deliberate act on the part of the plaintiff causing injury to the plaintiff then trespass would lie and the plaintiff should bring his action in that form.' 102 Similarly, Maxwell J held that 'in relation to a wilful act the plaintiff is confined to an action for trespass against the defendant for his own personal act'. 103

The action on the case for indirect harm was the forerunner of the modern law of negligence. By the nineteenth century, the law had come to recognise particular duties to take care in certain relationships that would be breached by an act or omission that caused harm to another inadvertently. These cases were 'directly opposed to intention where the mind fully appreciates the duty'. Following the abolition of the forms of action, 105 it took another 50 years before Lord Atkin articulated a general conception of a duty of care in tort in *Donoghue*, such that in *Grant v Australian Knitting Mills*, 106 it was said that 'the decision treats negligence, where there is a duty to take care, as a specific tort in itself'. The correlation that developed between negligence and careless or inadvertent injury, as contrasted with wilfully inflicted injury actionable in trespass, was evident in scholarly analysis. For example, in the first edition of *Salmond's Law of Torts*, published in 1907, the exclusion of intentionally inflicted injury from the scope of negligence was discussed as follows:

§5-Negligence

Negligence is the breach of a legal duty to take care. It is carelessness in a matter in which carefulness is made obligatory by law. Negligence and wrongful intent are the two alternative forms of *mens rea*, one or other of which is commonly required as a condition of liability. Each consists in a certain mental attitude of the defendant towards the consequences of his act. He intends those consequences when he foresees and desires them, and there does the act in order that they may happen. He is guilty of negligence, on the other hand, when he does not desire the

¹⁰¹ (1951) 51 SR (NSW) 179.

¹⁰² Ibid 182 (Street CJ, with whom Owen J agreed).

¹⁰³ Ibid, (Maxwell J, with whom Owen J also agreed).

¹⁰⁴ Percy H Winfield, 'The History of Negligence in the Law of Torts' (1926) 42(2) *Law Quarterly Review* 184, 184.

¹⁰⁵ Supreme Court of Judicature Act 1873, 36 & 37 Vict, c 66; Supreme Court of Judicature Act 1875, 38 & 39 Vict, c 77. See, eg, Winfield (n 104).

¹⁰⁶ (1935) 54 CLR 49 ('Australian Knitting Mills').

¹⁰⁷ Ibid 63 (Lord Wright for the Board).

consequences, and does not act in order to produce them, but is nevertheless indifferent or careless whether they happen or not, therefore does not refrain from the act notwithstanding the risk. The careless man is he who does not care — who is not anxious that his activities shall not be the cause of loss to others. The wilful wrongdoer is he who does not sufficiently desire to avoid doing it. Negligence and wrongful intent are inconsistent and mutually exclusive states of mind. He who causes a result intentionally cannot have caused it negligently, and vice versa.

...

The term negligence is used to denote not merely the mental attitude that has been described, but also conduct produced by such a mental attitude. In other words negligence means either subjectively a careless state of mind, or objectively careless conduct...This double use causes no difficulty or confusion, for negligence in the one sense is necessarily accompanied by negligence in the other also...Negligence in the subjective sense is opposed to wrongful intent; negligence in the objective sense is opposed to intentional wrongdoing.

It is negligence in its objective sense that is referred to in the well-known definition of Alderson B in *Blyth v Birmingham Waterworks Co*:¹⁰⁸ "Negligence is the omission to do something which a reasonable man, guided upon those considerations which ordinarily regulate the conduct of human affairs, would do, or doing something which a prudent and reasonable man would not do".¹⁰⁹

Understood this way, negligent conduct and intentionally inflicted injury are mutually exclusive. There is a purpose for tracing this early development of negligence by the courts and contemporaneous scholarly analysis. It suggests that the exclusion of intentionally inflicted harm was not a constraint imposed by forms of action, discarded with their abolition: it was a limitation that had shaped the scope of negligence as it developed an independent tort. Dias observed that:

The recognition of carelessness as a mode of inflicting harm has been variously traced by legal historians. When a sufficiently large volume of case-law has accumulated dealing with the careless infliction of harm, it is possible to draw a ring round them on paper, extract principles according to which remedies have been granted and refused, and then say that this is the "tort of negligence." ¹¹⁰

¹⁰⁸ (1856) 11 Ex 781, 784.

¹⁰⁹ Salmond (n 104) 18-20.

¹¹⁰ RWM Dias, 'The Duty Problem in Negligence' (1955) 13(2) Cambridge Law Journal 198, 199.

(c) 20th century developments

Many of the key 20th century cases were limitation cases, where the shorter limitation period for intentional torts was an obstacle sought to be overcome by pleading in negligence which attracted a longer limitation period. *Williams v Milotin* ('*Milotin*') concerned limitation periods in respect of injuries caused by negligent driving of a car by the defendant. The issue was whether the action for damages for personal injury was available in either of negligence (for which the limitation period was six years) or in trespass (for which the three-year limitation period had expired), at the election of the plaintiff. It was held that in the circumstances, it could have been brought in either and so was not out of time. However, in so finding, the majority distinguished the facts from intentional harm, saying '[t]here is no suggestion that the defendant intended to strike him. If that had been the allegation the action could have been brought in trespass and not otherwise.'¹¹¹ The court cited *Williams* as one of the authorities for the finding.

Following *Milotin*, another case arising from limitation issues under English law was decided by the Court of Appeal in *Letang v Cooper*.¹¹² On the scope of negligence, Lord Denning MR found that '[i]f [a person] does not inflict injury intentionally, but only unintentionally, the plaintiff has no cause of action today in trespass. His only cause of action is in negligence, and then only on proof of reasonable care.'¹¹³ Handford has rightly noted that Lord Denning MR did not say, in terms, that for *intentionally caused* injuries, the *only* cause of action was trespass.¹¹⁴ However, subsequently in *Gray v Barr*,¹¹⁵ Lord Denning MR held so unambiguously. The issue was whether Mr Barr's home insurance policy coverage of 'accidents' applied to the circumstances where he had gone into Mr Gray's house with a loaded gun (suspecting that Mr Gray was with Mr Barr's wife), firing one shot deliberately. During a physical fight between the men, the gun went off again, killing Mr Gray. Lord Denning MR held that the 'dominant and effective cause' of the shooting was the 'deliberate act in going up the stairs with a loaded shotgun',¹¹⁶ observing that:

If the death of Mr Gray was caused by the deliberate act of Mr Barr in going up the stairs with a loaded gun, it was no accident. ...The first question, therefore, is whether Mr Gray, had he lived, would have had a cause of action. The answer is: Yes, he would: but it would have been an action for assault, not for negligence. Whenever two men have a fight and

¹¹¹ Milotin (n 75) 470 (Dixon CJ, McTiernan, Williams, Webb and Kitto JJ).

^{112 [1965] 1} QB 232 ('Letang').

¹¹³ Ibid 239.

¹¹⁴ Handford (n 74) 41.

¹¹⁵ [1971] 2 QB 554 ('Barr'). ¹¹⁵ See also Trindade (n 82).

¹¹⁶ Barr (n 115) 567.

one is injured, the action is for assault, not for negligence. ...The idea of negligence — and contributory negligence — is quite foreign to men grappling in a struggle.¹¹⁷

To date, the High Court of Australia has continued to distinguish intentionally inflicted harm from negligence. In *Burnie Port Authority*, the majority of the High Court described negligence as 'the territory of tortious liability for *unintentional injury* to the person or property of another'¹¹⁸ and in *Northern Territory v Mengel* ('*Mengel*'),¹¹⁹ the majority distinguished negligence from 'torts which impose liability on private individuals for the intentional infliction of harm',¹²⁰ noting that 'the recent trend of legal development...has been to the effect that liability in tort depends on either the intentional or the negligent infliction of harm'.¹²¹ Four judges, Mason CJ, Dawson, Toohey and Gaudron JJ, were in the majority in both *Burnie Port Authority* and *Mengel*.

Twenty years before his appointment to the United Kingdom Supreme Court, Philip Sales (as he then was) and Daniel Stilitz cited these views of the Australian High Court in *Mengel*, amongst other judicial references,¹²² for the proposition that the law of tort could be 'organised at the two levels of intentional infliction of harm and failure to take reasonable care',¹²³ distinguishing 'harm inflicted intentionally, as a counterpart to the generalisation of the reasonable care or reasonable behaviour standard in the non-intentional sphere'.¹²⁴ The authors noted that the Australian High Court's observation in *Mengel* referred to above echoed the words of Lord Denning MR in *Letang* that claims for personal injury were now classified by whether 'the defendant did the injury intentionally or unintentionally'.¹²⁵

In their article, Sales and Stilitz were not addressing the non-delegable duty of care; rather, the scope of the tort of interference with trade or business by unlawful means. The authors argued that the tort would be more accurately described as 'intentional infliction of harm by unlawful means' and that 'a more helpful approach to understanding the limits of the tort lies in focusing more closely upon the element of intention'. But the authors' broader thesis remains relevant to this discussion.

¹¹⁷ Ibid 569.

¹¹⁸ Burnie Port Authority (n 1) 544 (Mason CJ, Deane, Dawson, Toohey and Gaudron JJ) (emphasis added).

^{119 (1995) 185} CLR 307 ('Mengel').

¹²⁰ Ibid 347 (Mason CJ, Dawson, Toohey, Gaudron and McHugh JJ).

¹²¹ Ibid 341.

¹²² Including *Hunter v Canary Wharf* [1997] AC 655, 707E-G (Lord Hoffmann).

¹²³ Sales and Stilitz (n 82) 436–7.

¹²⁴ Ibid 436.

¹²⁵ Letang (n 112) 239E-240B (Lord Denning MR).

¹²⁶ Sales and Stilitz (n 82) 411.

The authors argued for an organisation of torts such that 'the two principles of liability, for intentional infliction of harm and for failure to take reasonable care, should be regarded as base rules'. These 'base rules' are consistent with the statements in *Burnie Port Authority* and *Mengel* discussed above, and with the *ratio* of *Lepore* considered next.

(d) Lepore

Lepore concerned the scope of the non-delegable duty of care owed by the State to a child in a State school in respect of sexual abuse of the child by the schoolteacher. The New South Wales Court of Appeal had found that the non-delegable duty of care owed by the State could be breached by such intentional wrongdoing. In that case, a teacher was convicted of common assault of a child in punishing him for misbehaviour by sending him to a small room adjacent to the classroom, making him take his clothes off, and smacking the child on his bare buttocks and indecently handling him. It is noteworthy that the Court of Appeal regarded the question as novel. Mason P noted that '[i]t sometimes happens that a court is faced with a specific new situation in which there is no direct authority.' The majority held that the non-delegable duty was breached by the deliberate assault. Mason P held:

In my view the State's obligations to school pupils on school premises and during school hours extend to ensuring that they are not injured physically at the hands of an employed teacher (*whether acting negligently or intentionally*).¹³⁰

The same question arose in *Rich v State of Queensland*.¹³¹ The Queensland Court of Appeal unanimously refused to apply the majority's reasons in the New South Wales Court of Appeal in *Lepore* on this issue, agreeing with the reasons of Heydon JA in dissent. McPherson JA observed that:

Having regard to what I consider was decided in *Commonwealth v Introvigne*, the 'undertaking' [in a non-delegable duty], if that is what it really is, is not 'to ensure that no harm befalls the plaintiff', but at most an undertaking that the State would take reasonable steps to ensure the safety of persons like the plaintiffs. That is consistent with the decision in *Carmarthenshire County Council v Lewis* [1955] AC 549, from which the liability is derived.¹³²

¹²⁷ Ibid 437.

¹²⁸ Lepore (NSWCA) (n 25).

¹²⁹ Ibid 421–2 [1] (Mason P, Davies AJA agreeing at 448 [127]).

¹³⁰ Ibid 432 [61] (Mason P, Davies AJA agreeing at 448 [127]) (emphasis added).

¹³¹ (2001) Aust Torts Reports 81-626 (McPherson, Thomas and Williams JJA).

¹³² Ibid 67,392 [18].

Thomas and Williams JJA each observed that to give it such scope would introduce a new rule rendering the employer absolutely liable. Thus, this question was squarely in issue in the appeal to the High Court.

In the High Court, a majority also rejected the application of the non-delegable duty of care to intentionally inflicted harm. Gleeson CJ, with whom Callinan J agreed, observed that

although deliberately and criminally inflicting injury on another person involves a failure to take care of that person, it involves more. If a member of a hospital's staff with homicidal propensities were to attack and injure a patient, in circumstances where there was no fault on the part of the hospital authorities, or any other person for whose acts or omissions the hospital was vicariously responsible, the common law should not determine the question of the hospital's liability to the patient on the footing that the staff member had neglected to take reasonable care of the patient. It should face up to the fact that the staff member had criminally assaulted the patient, and address the problem of the circumstances in which an employer may be vicariously liable for the criminal acts of an employee. Intentional wrongdoing, especially intentional criminality, introduces a factor of legal relevance beyond a mere failure to take care. *Homicide, rape, and theft are all acts that are* inconsistent with care of person or property, but to characterise them as failure to take care, for the purpose of assigning tortious responsibility to a third party, would be to evade an issue. 133

Gleeson CJ's example of the act of a homicidal hospital staff member constituting a crime, not negligence, directly illustrated the question of the scope of negligence to apply to intentionally inflicted harm and it was rejected. Gummow and Hayne JJ similarly commented that all of the cases in which a non-delegable duty has been found have been cases of negligence, so to extend non-delegable duty to intentional acts would be to 'sever the duty from its roots in the law of negligence'. ¹³⁴ *Milotin* was applied as authority for the exclusion from negligence of intentionally caused harm. Their Honours said:

As *Williams v Milotin* makes plain, negligently inflicted injury to the person can, in at least some circumstances, be pleaded as trespass to the person, but the intentional infliction of harm cannot be pleaded as negligence. 135

_

¹³³ *Lepore* (n 4) 531–2 [31] (emphasis added).

¹³⁴ Ibid 602 [266] (Gummow and Hayne JJ).

¹³⁵ Ibid 602-3 [270].

This reasoning clearly demonstrates that the scope of the non-delegable duty was viewed by the Court through the lens of negligence, and that criminally intentional wrongdoing was regarded as lying beyond its focus.

However, McHugh J dissented on the issue. His Honour held that

a State education authority owes a duty to a pupil to take reasonable care to prevent harm to the pupil. The duty cannot be delegated. If, as is invariably the case, the State delegates the *performance* of the duty to a teacher, the State is liable if the teacher fails to take reasonable care to prevent harm to the pupil. The State is liable even if the teacher intentionally harms the pupil. The State cannot avoid liability by establishing that the teacher intentionally caused the harm even if the conduct of the teacher constitutes a criminal offence. It is the State's duty to protect the pupil, and the conduct of the teacher constitutes a breach of the State's own duty. 136

McHugh J reasoned further that an action for negligent infliction of harm is not barred by reason of the intentional act of the person causing the harm and that following abolition of the forms of action, a plaintiff may, if he or she chooses, sue in negligence for the intentional infliction of harm. ¹³⁷ McHugh J was the only judge in the Court who clearly accepted that negligence conceptually extends to intentionally caused harm.

Since *Lepore*, its *ratio* has been applied to preclude negligence from extending to intentionally inflicted harm. For example, *Nationwide News Pty Ltd v Naidu* concerned the liability of an employer for intentional conduct by way of racial slurs and vilification by one employee of another. Spigelman CJ noted that in respect of the wrongdoer's 'intentional course of conduct, which brutalised the respondent, it is artificial to analyse his conduct in terms of a duty of care owed by either appellant'. The Chief Justice observed that the imperial march of negligence had sometimes led to a 'lack of rigour' in the analysis of a cause of action, particularly in the context of differentiating negligence from intentional conduct, 139 citing both *Lepore* and *Milotin* for the proposition that 'negligently inflicted injury to the person can, in at least some circumstances, be pleaded as trespass to the person, but the intentional infliction of harm cannot be pleaded as negligence'. 140

¹³⁶ Ibid 562 [136] (emphasis added).

¹³⁷ Ibid 572 [162].

¹³⁸ Nationwide News (n 98) 484 [61].

¹³⁹ Ibid 484-5 [62].

¹⁴⁰ Ibid 485 [65].

There has been scholarly criticism of the *ratio* in *Lepore*. In particular, Robert Stevens has described the decision as 'indefensible' and argued that:

Liability for the breach of a primary duty cannot be avoided by showing that the breach was gross. ... If the duty assumed is a duty that care will be taken, this is breached where the child is abused. This does not mean that the duty assumed is absolute. A child who falls over in the playground does not necessarily have a claim. However, liability for deliberate abuse follows *a fortiori* from liability for want of care.¹⁴¹

This argument supports the reasons of McHugh J in *Lepore* that negligence is simply concerned with conduct that falls below the legal standard, however caused, and that following the abolition of forms of action, the gates of negligence have been opened to intentionally inflicted harm. However, as discussed above, writing exactly a century before Robert Stevens,142 Sir John Salmond had noted that attempts were sometimes made to define negligence as 'a purely objective fact in terms of unreasonable conduct or a failure to take care, regardless of the state of mind of the actor'. Salmond regarded this as a 'defective analysis of the conception'. This was because harm arising from a failure to take precautions may be 'either intentional wrongdoing or a mere accident'. Salmond opined that 'negligence in the subjective sense is opposed to wrongful intent; negligence in the objective sense is opposed to intentional wrongdoing'.144

Neither the abolition of forms of action, 145 nor the lack of express mens rea in the elements of the tort, provide a complete answer to the scope of negligence to apply to intentionally inflicted harm under Australian law. Its historical foundation in inadvertent harm seems to have shaped its development as a cause of action. Its elements set the parameters for when a duty will arise and when it will be breached, but the underlying restriction to unintentional harm appears to derive from precedent. It is argued here that the ratio in Lepore has not been significantly challenged and remains good law. There are subsequent observations of the Court consistent with the limitation. For example, following Lepore, in Magill v Magill, Gleeson CJ again discussed the duty of care in negligence as controlling 'potential liability for carelessness'. His Honour expressly distinguished between false representations (in that case about paternity) which could be the result of

¹⁴² See Salmond (n 104) 19–21. Cf Robert Stevens, *Torts and Rights* (Oxford University Press, 2007)

¹⁴³ Salmond (n 104) 20.

¹⁴⁴ Ibid.

¹⁴⁵ Above n 105.

'carelessness' or "deliberate fraud'. 146 Gummow, Kirby and Crennan JJ also referred to negligence as an 'unintentional tort'. 147 Another way in which the limitation to unintentional harm or careless conduct seems to find expression is in the notion of what is a relevantly foreseeable risk. For example, in *Minister for the Environment v Sharma*, 148 Beach J observed that:

The question posed by the consideration of foreseeability at the stage of considering whether a duty of care exists is whether 'it is reasonably foreseeable as a possibility that *careless conduct* of any kind on the part of the defendant may result in damage of some kind to the person or property of the plaintiff'...¹⁴⁹

Similarly, in *Sullivan v Moody*,¹⁵⁰ the majority observed:

But the fact that it is foreseeable, in the sense of being a real and not farfetched possibility, that a careless act or omission on the part of one person may cause harm to another does not mean that the first person is subject to a legal liability to compensate the second by way of damages for negligence if there is such carelessness, and harm results. ...A defendant will only be liable, in negligence, for failure to take reasonable care to prevent a certain kind of foreseeable harm to a plaintiff, in circumstances where the law imposes a duty to take such care.¹⁵¹

However, the resolution to the debate as to the scope of negligence and the non-delegable duty lies with the High Court. For present purposes, the main point is that the *ratio* of *Lepore* was clear, and to change the law will require overturning that case.

(e) Limited judicial support for negligence to apply to intentionally caused harm

In Australian Securities and Investments Commission v Cassimatis (No 8) ('Cassimatis'),¹⁵² Edelman J said that, '[i]n Australia, an action for the tort of negligence can succeed even where the conduct of the defendant was intentional.'¹⁵³ Assuming his Honour was referring to intentionally inflicted harm, rather than merely voluntary conduct, the authorities cited for the proposition need to be

¹⁴⁶ Magill (n 98) 563 [21] (Gleeson CJ).

¹⁴⁷ Ibid 589 [117].

^{148 (2022) 291} FCR 311 ('Sharma').

¹⁴⁹ Ibid 423 [417] (Beach J) (emphasis added).

¹⁵⁰ (2001) 207 CLR 562.

¹⁵¹ Ibid 576 [42] (Gleeson CJ, Gaudron, McHugh, Hayne and Callinan JJ) (emphasis added).

^{152 (2016) 336} ALR 209 ('Cassimatis').

¹⁵³ Ibid 306 [504] (Edelman J).

considered. These were *Milotin*;¹⁵⁴ *Wilson v Horne* ('*Wilson*');¹⁵⁵ *Gray v Motor Accident Commission* ('*Gray*');¹⁵⁶ and *Croucher v Cachia* ('*Croucher*').¹⁵⁷ They are considered below. It is respectfully suggested that with the exception of *Wilson* (which clearly supported the extension of negligence to intentionally inflicted harm in the form of sexual abuse, but which pre-dated *Lepore*), the cases do not lend strong support to the proposition that intentionally inflicted injury is properly characterised or pleaded as negligence under Australian law.

As discussed above, *Milotin* is authority for the fact that direct contact may amount to trespass or negligence under Australian law, but that intentional harm cannot be pleaded as negligence. Even if rationalised as a case dealing with pre-judicature reforms, this does not directly support the reach of negligence to intentionally caused harm. The abolition of forms of action did not affect the substantive elements of causes of action.¹⁵⁸

*Gray*¹⁵⁹ was decided by the High Court five years before *Lepore*. The appellant (the plaintiff at trial) was struck by a car intentionally driven at him by a person who was convicted of causing grievous bodily harm with intent to cause such harm. In 1993, the appellant commenced proceedings against the wrongdoer in the District Court of South Australia, claiming damages for personal injury. As noted by the High Court, the District Court action was framed as a claim in negligence, but appeared to be conducted as a claim in trespass. The respondent, who was the compulsory third party insurer of the wrongdoer, was substituted as defendant in the action pursuant to provisions of the *Motor Vehicles Act 1959* (SA). Although the respondent did not admit liability, the High Court noted there seems to have been no real dispute about liability at trial. Accordingly, the finding of negligence in the District Court arising from intentionally caused harm does not appear to have been the subject of argument at trial or in the Court of Appeal or High Court.

¹⁵⁴ *Milotin* (n 75).

^{155 (1999) 8} Tas R 363 ('Wilson').

¹⁵⁶ Gray (n 75).

^{157 (2016) 95} NSWLR 117 ('Croucher').

¹⁵⁸ As noted by Edelman J in *H Lundbeck A/S v Sandoz Pty Ltd* (2022) 276 CLR 170, 212 [117]: "The most common use of "cause of action" under Australian law is in the sense set out in *Letang v Cooper*, where Diplock LJ commenced his reasons with the description of a cause of action as "simply a factual situation the existence of which entitles one person to obtain from the court a remedy against another person".

¹⁵⁹ *Gray* (n 75).

¹⁶⁰ Lepore (n 4).

¹⁶¹ Gray (n 75) 9-10 [24].

The issue was the availability of exemplary damages in relation to negligence. As discussed above in the section addressing intention, it had already been accepted before Gray's case that highhanded conduct in contumelious disregard of the rights of a plaintiff could give rise to exemplary damages in negligence. The cases cited by the majority in $Gray^{162}$ were cases where an employer had acted consciously with respect to the conditions in the workplace, with a high-handed disregard for the health of the workers and exposure to harmful substances such as asbestos or benzol fumes preferencing corporate profits over installing safety equipment. However, in none of those case was the defendant found to have intended to injure the plaintiff, nor was the harm direct.

In the joint judgment of Gleeson CJ, McHugh, Gummow and Hayne JJ, their Honours observed that

exemplary damages could not properly be awarded in a case of alleged negligence in which there was no conscious wrongdoing by the defendant. Ordinarily, then, questions of exemplary damages will not arise in most negligence cases be they motor accident or other kinds of case. But there can be cases, framed in negligence, in which the defendant can be shown to have acted consciously in contumelious disregard of the rights of the plaintiff or persons in the position of the plaintiff. Cases of an employer's failure to provide a safe system of work for employees in which it is demonstrated that the employer, well knowing of an extreme danger thus created, persisted in employing the unsafe system might, perhaps, be of that latter kind. No doubt other examples can be found.¹⁶³

This passage has been considered difficult to reconcile with a limitation of negligence as not applying to intentionally inflicted harm because of the Court's acceptance that exemplary damages were available in these circumstances of deliberate wrongdoing. One the one hand, the Court expressed no criticism of the finding of negligence below, but on the other, it was not in issue. Given that the finding of negligence was not argued at any stage in the litigation, the case cannot be regarded as authority for the proposition that negligence applies to intentionally inflicted harm. A case is not authority for a proposition that was not argued, or was simply agreed or assumed. Accordingly, *Gray* provides limited support at best for the scope of negligence to apply to intentionally inflicted harm.

¹⁶² Ibid 27 [40], citing *Midalco Pty Ltd v Rabenalt* [1989] VR 461, 29 [86], *Trend Management* (n 85); *Coloca v BP Australia Ltd* [1992] 2 VR 441. See also Kirby J at [82].

¹⁶³ Gray (n 75) 9-10 [22].

¹⁶⁴ CSR Ltd v Eddy (2005) 226 CLR 1, 11 [13] (Gleeson CJ, Gummow and Heydon JJ).

Wilson¹⁶⁵ was a decision of the Full Court of the Supreme Court of Tasmania, decided the year after *Gray*, and before *Lepore*. This case directly supported the scope of negligence to apply to intentionally caused harm in the form of sexual abuse. It was also a case in which the effect of then existing limitation periods loomed large in the disposition of the case. The cause of action in trespass had expired, but an action in negligence remained available. The Court found that sexual abuse of the plaintiff as a child by her uncle could be pleaded as negligence. Cox CJ limited the relevance of Milotin to pre-judicature reforms. His Honour cited Gray as a (then) recent 'example of conduct involving an intentional trespass to the plaintiff's person but pleaded in negligence',166 noting that it was 'implicit' 'that the court at first instance had jurisdiction to entertain an action for damages based on negligence as the pleaded cause of action, notwithstanding that the tortfeasor's conduct might also have given rise to a cause of action in trespass'. 167 However, his Honour did not discuss the fact that the issue was agreed at trial in the District Court and not the subject of appeal in the High Court. Wright I similarly regarded *Milotin* as no bar to the claim. 168 Evans J took a similar view, and also noted that the trial judge had cited from the ninth edition of Charlesworth & Percy's The Law of Negligence (1997) for the proposition that '[n]egligence in this context may consist in a wilful or an intentional act.'169

As discussed above, there is no doubt that negligence can consist of an intentional act. The uncertain question is whether it can arise from intentionally inflicted harm. In the 16th edition of Charlesworth and Percy,¹⁷⁰ which was in almost identical terms to the edition cited in *Wilson*, the editors state that '[n]egligence may consist in a wilful or an intentional act. This is because negligence as the breach of a duty to take care is concerned with conduct and not with intention.'¹⁷¹ However, the sole authority cited for the proposition is a somewhat unusual case, *Emblen v Myers*.¹⁷²

¹⁶⁵ *Wilson* (n 155).

¹⁶⁶ Ibid 367 [7].

¹⁶⁷ Ibid.

¹⁶⁸ Ibid 373-4 [21].

¹⁶⁹ Ibid 382 [40].

 $^{^{170}}$ Mark Armitage et al (eds), *Charlesworth & Percy on Negligence* (Sweet & Maxwell, 16^{th} ed, 2025) 12-13.

¹⁷¹ John Charlesworth, *The Law of Negligence*, (Sweet & Maxwell, 2nd ed, 1938) 9. I note that for the purposes of this discussion, intentional conduct is discussed in terms of the deliberate act intending to cause the consequences, rather than simply in terms of a voluntary act. See Cane (n 82) 35.

¹⁷² Emblen v Myers (1860) 6 H & N 54; 158 ER 23. The plaintiff owned land with a stable. The defendant owned an adjoining house that had been ordered to be demolished. The plaintiff refused to sell his property to the defendant. The defendant's workmen subsequently recklessly demolished the building, causing a beam to fall on the plaintiff's property. Following an argument, the defendant instructed his workmen to continue, and they threw brick onto the stable. During the course of the trial, the defendant's counsel did not object to evidence in a claim in trespass being introduced in the

While the case supports the availability of exemplary damages in negligence, in *Rookes v Barnard*¹⁷³ Lord Devlin described the case as one in trespass, and one of six early cases dealing with exemplary damages without 'any clear ruling'. Wilson clearly accepted that negligence extended to intentionally inflicted harm. However, it must be regarded as overruled following *Lepore*.

Croucher¹⁷⁵ is the final case to which Edelman J referred to in *Cassimatis* as supporting the proposition that negligence can apply to intentional conduct. The case is relevant to understanding the divergence between Australian and English law insofar as English law¹⁷⁶ does not allow unintentional, direct contact to be pleaded in trespass, whereas Australian law does. Although battery is classed as an 'intentional tort', a battery may occur when the defendant is merely negligent, so that depending on the circumstances, 'the actual facts will or may fulfil the requirements of [battery and negligence]'.¹⁷⁷ However, it is respectfully suggested that *Croucher* is not authority for the inverse proposition that intentionally inflicted harm can be pleaded as negligence.

(f) The position under English law in relation to the non-delegable duty

It is noteworthy that in a decision of the United Kingdom Supreme Court there is *obiter* to the effect that a non-delegable duty of care *can* be breached by the intentional criminal wrongdoing of a delegate. In *Armes v Nottingham Shire County Council* ('Armes'), ¹⁷⁸ Lord Reed JSC observed:

Nor am I able to agree that a non-delegable duty cannot be breached by a deliberate wrong: see, for example, *Morris v C W Martin & Sons Ltd* [1966] 1 QB 716, a bailment case which was treated as a case of non-delegable duty in *Woodland's* case [2014] AC 537, 573–574, para 7. On Burnett LJ's approach, the local authority would seemingly be liable if the foster parents negligently enabled a third party to abuse the child, but not if they abused her themselves. That can hardly be right.¹⁷⁹

action that had been brought in negligence. The trial judge allowed the jury to award exemplary damages in the negligence action if they found the defendant had wilfully attempted to drive the plaintiff from his stable.

¹⁷³ [1964] AC 1129.

¹⁷⁴ Ibid 1223. Lord Devlin noted that Wilde B at trial had directed the jury that they might find exemplary damages 'without recourse to legal principle'.

¹⁷⁵ Above n 157.

¹⁷⁶ See, eg, *Letang* (n 112).

¹⁷⁷ Croucher (n 157) 121 [22] (Leeming JA, Beazley P agreeing at 118–19 [1], Ward JA agreeing at 119 [2]).

¹⁷⁸ Armes (n 23).

¹⁷⁹ Ibid 375 [51] (Lord Reed, with whom Baroness Hale PSC, Lord Kerr and Lord Clarke JSC agreed).

However, if the real reason that there is no common law liability for intentional wrongdoing by independent contractors lies in the limitation of vicarious liability, and the refusal of the courts to extend to independent contractors under any circumstances, it is not clear why 'it can hardly be right' that a non-delegable duty does not apply to intentional wrongdoing. Both the limitations of vicarious liability, and the limited scope of the non-delegable duty of care in negligence, reflect the limitations of separate doctrines that serve different normative purposes. It is respectfully submitted that an unacceptable gap in the law is a separate question, and may properly be one for law reform.

The most detailed consideration of the scope of the non-delegable duty in the United Kingdom was by Lord Sumption JSC in *Woodland v Swimming Teachers Association* ('*Woodland*').¹⁸⁰ *Woodland* was not a case addressing intentional wrongdoing. The facts concerned a child who suffered a hypoxic brain injury during swimming lessons at a local pool. The school was found liable under a non-delegable duty of care. Lord Sumption carefully reviewed the authorities, including from Australia — and specifically *Lepore* — and observed that they 'are broadly correct in their analysis of the factors that have given rise to non-delegable duties of care'. ¹⁸¹ If his Lordship disagreed with the express reasons of the majority in *Lepore* in restricting the scope of the non-delegable duty to negligent and not intentionally caused harm, that would have been the place to say it, even in *obiter*. His Lordship identified five elements of the non-delegable duty, but also noted that:

This is not an open-ended liability, for there are important limitations on the range of matters for which a school or education authority assumes non-delegable duties. They are liable for the negligence of independent contractors only if and so far as the latter are performing functions which the school has assumed for itself a duty to perform, generally in school hours and on school premises (or at other times or places where the school may carry out its educational functions). In the absence of negligence of their own, for example in the selection of contractors, they will not be liable for the negligence of independent contractors where on

¹⁸⁰ Woodland (n 63).

¹⁸¹ Ibid 583 [23]. In *Woodland*, Lord Sumption identified two broad categories of case in which a non-delegable duty of care has been held to arise. The first was 'a large, varied and anomalous class of cases in which the defendant employs an independent contractor to perform some function which is either inherently hazardous or liable to become so in the course of his work'. The second broad category was said to comprise cases where the common law imposed a duty which had three critical characteristics. First, the duty arises because of an antecedent relationship between the defendant and the claimant. Secondly, the duty is a positive or affirmative duty to protect a particular class of persons against a particular class of risks, and not simply a duty to refrain from acting in a way that foreseeably causes injury. Thirdly, the duty is by virtue of that relationship personal to the defendant.

analysis their own duty is not to perform the relevant function but only to arrange for its performance. 182

It is argued here that Lord Sumption JSC's reasons do not confirm or necessarily even support the reach of the non-delegable duty to criminally intentional wrongdoing. It is beyond the scope of this article to consider the position under the law of England and Wales in detail, but it would appear that the scope of the non-delegable duty as suggested in *obiter* in *Armes* is inconsistent with the ratio of the Australian High Court in *Lepore*. It is to be hoped that the United Kingdom Supreme Court will also have an opportunity to clarify the scope of the non-delegable duty and its relationship with negligence, as well as address the anterior question of the scope of negligence to apply to intentionally inflicted injury.

V FACTORS RELEVANT TO OVERTURNING LEPORE

The application to overturn the earlier decision of the Court in *Lepore* will fall to be determined in accordance with the settled principles in invoking the authority of the High Court to reconsider a previous decision. Although there is 'no very definite rule' as to the circumstances in which the High Court will do so, there are a number of recognised considerations which will be relevant. These include whether the earlier decision 'did not rest upon a principle carefully worked out in a significant succession of cases'; differences in the reasons of the justices constituting the majority; whether the earlier decisions had 'achieved no useful result but on the contrary had led to considerable inconvenience'; and whether the earlier decision had been independently acted on in a manner which militated against reconsideration. The earlier decision is to be treated as 'the primary premise from which other arguments follow, not just as one potential premise amongst an aggregate of competing premises', and the questions will be informed by 'a strongly conservative cautionary principle, adopted in the interests of continuity and consistency in the law'. 186

As discussed above, there was no difference of substance in the reasons of the majority in *Lepore* — each of the judgments in the majority proceeded on the basis that the metes and bounds of the non-delegable duty were determined by the law of

¹⁸² Ibid 584-5 [23], [25(3)].

¹⁸³ See *John v Federal Commissioner of Taxation* (1989) 166 CLR 417, 438–9 (Mason CJ, Wilson, Dawson, Toohey and Gaudron JJ).

¹⁸⁴ Ibid.

¹⁸⁵ Lee v New South Wales Crime Commission (2013) 251 CLR 196, 232–3 [66] (Hayne J), quoting Bryan Horrigan, 'Towards a Jurisprudence of High Court Overruling' (1992) 66(4) Australian Law Journal 199, 208.

¹⁸⁶ Wurridjal v Commonwealth (2009) 237 CLR 309, 352 [70] (French CJ).

negligence, and that to allow it to be breached by intentionally caused harm would 'sever' the connection with negligence. Their analysis of the scope of negligence was consistent with the earlier decision in *Burnie Port Authority* maintaining 'a coherent law of negligence to dominate the territory of tortious liability for unintentional injury to the person or property of another'.¹⁸⁷

There are several other reasons for caution in considering this significant development of the non-delegable duty at common law and overturning *Lepore*. The first is that a significant expansion of the non-delegable duty at common law would render the carefully worked out principles of vicarious liability redundant in any relationship to which a non-delegable duty applies. As noted in *Lepore*, 188 the non-delegable duty of care at common law has no comparable in-built limitations in the form of a requirement for a sufficient connection between the wrongdoing and the duties assigned to the worker as apply under the doctrine of vicarious liability. The effect of an extension of the non-delegable duty to undercut the operation of the doctrine of vicarious liability is a significant factor to take into consideration. 189 The common law is inherently cautious about the development of one area of the law in a manner that would overtake or eclipse another. There is no question that the law recognises that a set of facts may give rise to overlapping claims. 190 However, that is quite a different matter to the development of the law in a manner that overrides or is inconsistent with another part of the law. In Breen, Gaudron and McHugh JJ cautioned that '[a]ny changes in legal doctrine, brought about by judicial creativity, must "fit" within the body of accepted rules and principles.'191 More recently in Sharma, 192 Allsop CJ discussed the concept of coherence in development of the law and observed that, '[i]n its broadest sense, coherence pertains to the consistency and harmony (logically and normatively) of the legal system as a whole'. 193 In Perera, 194 Leeming JA, with whom Macfarlan and Simpson JJA agreed, observed that 'the law of

¹⁸⁷ Burnie Port Authority (n 1) 544-5 (Mason CJ, Deane, Dawson, Toohey and Gaudron JJ).

¹⁸⁸ *Lepore* (n 4) 532 [32] (Gleeson CJ).

¹⁸⁹ See further Kit Barker, 'Private Law as a Complex System: Agendas for the Twenty-First Century' in Kit Barker, Karen Fairweather and Ross Grantham (eds), *Private Law in the 21st Century* (Hart Publishing, 2017) 3, 3.

¹⁹⁰ Fistar v Riverwood Legion & Community Club Ltd (2016) 91 NSWLR 732, 743 [48]–[50] (Leeming JA).

¹⁹¹ Breen (n 15) 115.

¹⁹² Sharma (n 148).

¹⁹³ Ibid 392 [244], citing *Hill v Van Erp* (1997) 188 CLR 159, 231 (Gummow J). See also Craig Purshouse, 'Liability for lost autonomy in negligence: Undermining the coherence of tort law?' (2015) 22 *Torts Law Journal* 226, where the author discusses the importance of coherence in the tort of negligence in the context of whether tort law could recognise lost autonomy as actionable in negligence. See also Andrew Fell, 'The Concept of Coherence in Australian Private Law' (2018) 41(3) *Melbourne University Law Review* 1160.

¹⁹⁴ Perera (n 15).

negligence is but one part of the Australian legal system, and is not to be extended so as to obliterate or undercut other principles, which may serve other important values'. 195

Further, as noted by Gummow and Hayne JJ in *Lepore*, an extension to render the non-delegable duty a wider form of strict liability for intentionally inflicted harm would be a 'step sharply at odds with the trend of decisions in this Court rejecting the expansion of strict liabilities'. ¹⁹⁶

VI CONCLUSION

The development of negligence sought in the appeal poses significant questions for the High Court's consideration and the future direction of the common law. The distinct roles for negligence, vicarious liability and the non-delegable duty of care need to be very carefully balanced to preserve the coherence of the law as a whole. This article has argued that the non-delegable duty of care under Australian law is a clearly developed duty of care owed in negligence, albeit one that cannot be discharged by selection of an appropriately qualified contractor. The corollary is that the law of negligence should determine the parameters of the duty. It has also been argued that, despite some uncertainty and inconsistency in the cases, duties of care under the law of negligence remain concerned with carelessly caused harm and not intentionally caused harm. To overturn this *ratio* of *Lepore* could have profound implications for the law of negligence beyond non-delegable duties.

It may be that the appeal in *AA* can be determined without the Court expressing a concluded view on the ambit of the non-delegable duty, for example, if the trial judge's findings on the Diocese's direct liability in negligence are restored. On one view, that would accord with 'the standard common law judicial technique of deciding no more than what needs to be decided'. However, in light of the uncertainty surrounding the non-delegable duty, the appeal offers a valuable chance 'for some guidance to be provided by [the] Court...so as to reduce the risk of unnecessary appellate processes arising out of the existing uncertainties', 198 as the High Court was prepared to do in *Prince Alfred College Inc v ADC* 199 in relation to vicarious liability. It is to be hoped this opportunity is taken.

¹⁹⁵ Ibid 93 [42].

¹⁹⁶ Lepore (n 4) 601-2 [266].

¹⁹⁷ Mann v Paterson Constructions Pty Ltd (2019) 267 CLR 560, 597 [76] (Gageler J).

¹⁹⁸ Prince Alfred College (n 10) 143 [10] (French CJ, Kiefel, Bell, Keane and Nettle JJ).

¹⁹⁹ Prince Alfred College (n 10).

Finally, even if *Lepore* is upheld, this does not mean that the law should remain unchanged. It is well within the purview of the legislature to consider targeted legislative reform to render institutions that provide care to vulnerable children strictly liable for child abuse, whether by an employee or independent contractor. As observed by Stapleton, Parliament has the resources to carefully explore social and economic ramifications of reform, and reforms in controversial matters have the legitimacy that flows from an electoral mandate.²⁰⁰ Legislative reform may offer a means to address this sadly pervasive social problem in a targeted way rather than through doctrinal development by the courts.

²⁰⁰ Stapleton (n 13) 16.